Multidisciplinary : Rapid Review : Open Access Journal

Received 22 March 2023, accepted 18 April 2023, date of publication 24 April 2023, date of current version 28 April 2023. Digital Object Identifier 10.1109/ACCESS.2023.3270100

RESEARCH ARTICLE

A Research on Blockchain Technology: Urban Intelligent Transportation Systems in Developing Countries

XINGCHENG GUO^(D), (Member, IEEE), AND XIANGLONG GUO²

¹School of Management, Guangzhou City University of Technology, Guangzhou 510800, China ²Industrial Development Strategy Research Institute, Nanchang 330008, China

Corresponding author: Xingcheng Guo (guoxc2@gcu.edu.cn)

This work was supported in part by the University Online Course Committee Fund, Guangdong, China, under Grant 2022ZXKC589; and in part by the Private Higher Education Committee of Higher Education Association, Guangdong, under Grant 2022MBGJ044.

ABSTRACT Blockchain technology has been widely used in finance, transportation, education, medical treatment, network security, management science, and other industries due to its decentralization, high reliability, and traceability characteristics. Unlike other studies on Urban Intelligent Transportation Systems (UITS) in the past, we present a model framework of the Urban Intelligent Transportation Systems (UITS) applied by blockchain in developing countries. After a detailed elaboration of the situation of three representative Urban Intelligent Transportation Systems (UITS) in China, blockchain technology has been applied to build a new architecture model of the big data platform for urban intelligent Transportation systems (UITS) in developing countries. Finally, this paper elaborates on the critical direction of future development, areas of concern, and open research challenges, which researchers and urban intelligent transportation designers could explore to further advance in this field.

INDEX TERMS Blockchain, blockchain architecture, intelligent transport systems (ITS), urban intelligent transport systems (UITS), developing countries, security.

I. INTRODUCTION

The core technology of blockchain began to appear from the late 1980s to the early 1990s [1]. Satoshi Nakamoto proposed the concept of Bitcoin based on the core technology of blockchain [2]. As an emerging technology, blockchain has carried out applicability research based on its characteristics of decentralization, high reliability, and traceability. The blockchain is a universally acclaimed innovation based on distributed ledger technology [3]. Melanie Swan defined the development of blockchain technology at three levels [4]: blockchain 1.0 (currency interconnection),

mainly bitcoin and other virtual currencies; Blockchain 2.0 (asset interconnection) takes smart contracts as the primary feature and can develop decentralized application DAPP as required; Blockchain 3.0 (Internet of Things), the application

The associate editor coordinating the review of this manuscript and approving it for publication was Mansoor Ahmed^(D).

field of blockchain technology extends to education, government management, medical care, transportation, etc. It can confirm, measure and store the property rights of the valuable information in the blockchain and quickly establish credit, making the work more convenient and efficient [5].

The application of blockchain in transportation is crucial because public blockchain technology has become increasingly important. Transportation is considered the backbone of cities. It provides essential mobility services for people's daily life and logistics goods [6], [7], [8]. The gradual automation, digitalization, and intellectualization of transportation is the inevitable trend of transportation development. Apply intelligent technology to the transportation field to achieve the development goal of networked vehicles [9]. The story of intelligent transportation cannot be separated from the innovation of advanced technologies such as big data analysis, artificial intelligence, and information technology. However, integrating these emerging technologies has also

This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/ brought a certain degree of security risk to Intelligent Transportation Systems and caused social attention to the network security of Intelligent Transportation Systems. Currently, the cross-regional sharing mechanism in the Intelligent Transportation Systems network is not perfect, the cooperation efficiency between departments is low, and there are a series of network security problems such as massive data sharing and storage, information acquisition and information acquisition and tampering, and user privacy disclosure [10].

Blockchain technology is introduced into the intelligent transportation network to solve the above problems. The data under blockchain technology is authentic and can not be tampered with. It can provide technical support for storing, sharing, traceability, and anti-attack traffic data. Blockchain offers technical support for the construction of the transportation field, such as using blockchain technology to share traffic congestion, road traffic flow, and other data information to solve the problem of information blocking and data integration difficulties in transportation, combined with edge computing and other technologies, it is applied to the Internet of Vehicles to improve the overall safety and synergy of the system and provide safety guarantee for transportation. At present, Intelligent Transportation Systems (ITS) have become one of the key topics of academic research and also in some international standardization organizations, such as the Internet Engineering Task Force (IETF) [11] and International Standards Organization (ISO). The success of blockchain technology is conducive to solving problems such as network security of Intelligent Transportation Systems (ITS). ITS can be traced back to the 1970s. It is the direction of the future traffic system [12]. The essential underlying technologies of Intelligent Transportation Systems (ITS) include electronic sensing technology, intelligent control technology, data transmission technology, etc. [13], [14], [15] ITS aims to provide more convenient and safer travel and transportation modes for drivers, riders, and pedestrians.

Intelligent Transportation Systems (ITS) have attained huge fame. However, problems and new challenges in developing countries are still facing, especially when combined with blockchain technology, such as open issues in security and privacy.

A. CONTRIBUTION AND COMPARISON WITH RELATED RESEARCH ARTICLES

This paper mainly contributes to the existing knowledge in three broad ways. First, this paper studies the Urban Intelligent Transportation Systems (UITS) of China, a representative country in the developing countries, focusing on the current structural models of Intelligent Transportation Systems in Guangzhou, Shenzhen, and Foshan, analyzes the advantages and disadvantages of the structural models of Intelligent Transportation Systems in the three cities. It is conducive to the overall understanding and reference of other developing countries, especially for government officials in developing countries with relatively weak professional skills. Second, a new network architecture of the big data platform for urban intelligent transportation has been built based on blockchain technology. By taking advantage of the advantages of blockchain, such as decentralization, distributed computing, big data sharing, and incentive mechanism, the problem of independent management and transmission of data by various organizations of the traditional urban intelligent transportation data network architecture has been solved, thus promoting data sharing. Third, improve Urban Intelligent Transportation Systems (UITS) in developing countries, and propose a top-level design architecture model based on blockchain traffic cloud. However, based on the literature review, no paper has addressed detailed aspects of improving urban transportation intelligent systems (UITS) in developing countries based on blockchain technology. This lack of comprehensiveness motivates us to contribute through this research on the current situation of Urban Intelligent Transportation Systems (UITS) in developing countries and the future development of Intelligent Transportation Systems based on blockchain technology in developing countries.

Many survey articles have been written in the recent past with only a focus on specific ITS technology, application development, key success factors, etc. Some papers have also presented the applications of the traditional ITS model, etc. Strengthening the data security of ITS is also studied in some papers, including [16], [17], [18], [19], [20].

However, on the other side, the contribution of this research is as follows: To study the Urban Intelligent Transportation Systems (UITS) in developing countries, select the Intelligent Transportation Systems of three representative cities in China as the research objects, including the research system architecture, functions, advantages, and disadvantages. Moreover, a new network model architecture of the big data platform for urban intelligent transportation has been built based on blockchain technology. Finally, optimize the structure of Urban Intelligent Transportation Systems (UITS) in developing countries, and build a top-level model framework based on blockchain traffic cloud.

Ultimately the goal of this research is to acquaint researchers and urban transport system designers in developing countries with the urban intelligent transport system architecture based on blockchain technology and the detailed intelligent transportation system architecture model of major cities in China, a typical representative of developing countries.

B. ORGANIZATION OF THE SURVEY

The organization of the survey is as follows and shown in Figure 1. Section II discusses the background concepts as preliminaries to understand better this paper's core, followed by Blockchain and Intelligent Transportation Systems (ITS) characteristics. This section also highlights Blockchain technology and Intelligent Transportation Systems (ITS) challenges and issues. Section III presents the structure model of Intelligent Transportation Systems (ITS) of the three cities in China, a typical representative of developing countries in detail. In section IV, we propose a big data platform structure



FIGURE 1. Structure and contribution of this research.

for urban intelligent transportation based on blockchain technology. The research and innovation in constructing a new generation of urban intelligent transportation brain systems based on blockchain technology are highlighted in section V. Section VI describes the open research issues learned from this research and the literature review to carry out in the future, and finally, Section VII concludes the research.

II. BACKGROUND AND CHALLENGES

This section first presents the basic concepts of Blockchain Technology and Intelligent Transportation Systems (ITS) then discuss the characteristics and issues of the status of Urban Intelligent Transportation Systems(UITS) in developing countries.

A. BACKGROUND AND CHALLENGES OF BLOCKCHAIN TECHNOLOGY

Blockchain is an innovation mode and application of distributed data storage technology, P2P network, encryption algorithm, smart contract, consensus mechanisms, and other information technologies in the Internet era [21]. The birth of blockchain cannot be separated from Bitcoin. The Bitcoin system has been the most influential and well-known application case of blockchain technology. Broadly speaking, blockchain technology is a new distributed infrastructure and computing paradigm that uses blockchain data structure to verify and store data, uses distributed node consensus algorithm to generate and update data, uses cryptography to ensure the security of data transmission and access, and



FIGURE 2. Blockchain structure diagram.

uses smart contracts composed of automated script code to program and operate data [22]. Each block contains the data to be recorded, the current block of the root Hash, the previous block of the root Hash, the timestamp, and other information. The blockchain is formed by linking the root hash values of the front and back blocks according to the timestamp sequence [23], [24], [25]. The structure of the Blockchain is as follows and shown in Figure 2.

Blockchain has the characteristics of decentralization, sequential data, collective maintenance, programmability, security, and anonymity [26]. The various operation processes of blockchain data are based on the distributed system structure. Pure mathematical methods are used instead of the central mechanism to establish the trust relationship of each node, forming a decentralized distributed system. Each block has a timestamp, which adds a time attribute to the data to achieve data verification and traceability [27]. Each distributed node jointly participates in data maintenance and verification under the specific incentive mechanism and determines the specific new block into the chain through a consensus algorithm. The blockchain system provides a flexible script code system to support users to in creating smart contracts or other DAPPs according to their needs. The encryption is based on the principle of asymmetric cryptography, and the security of the system is guaranteed with the help of the consensus mechanism of the distributed system [28]. The transaction process does not need to pass the public identity, which has the characteristics of anonymity.

In recent years, blockchain has developed rapidly, and its application scenarios have quickly grown in other fields besides the financial area [29]. The United Nations, the United States, the United Kingdom, Japan, the IMF, multinational banks, securities companies, and other countries or institutions actively explore and promote the research and application of blockchain technology [30].

As a distributed ledger technology, blockchain overcomes the disadvantages of traditional centralized ledgers, such as low storage efficiency, low credibility, and vulnerability to single-point attacks. The technology ensures distributed sharing, multi-party consensus, tamper resistance, and traceability. Distributed ledger (a shared ledger or distributed ledger technology) is a replication consensus that shares and synchronizes digital data geographically distributed across multiple websites, countries, or institutions. No central administrator or centralized data storage [31].

At present, there are still many problems in the practical application of blockchain technology [32], [33]. For example, the distributed storage architecture has high hardware requirements for each physical node, the blockchain system throughput is small, the transaction information processing rate is low, the consensus mechanism still has vulnerabilities, and the smart contract may be attacked. These problems still restrict the application width of blockchain technology in practical scenarios, and the key to solving these problems is to build a good blockchain development environment in all fields [34].

B. BACKGROUND AND CHALLENGES OF URBAN INTELLIGENT TRANSPORTATION SYSTEMS(UITS)

Intelligent Transportation Systems (ITS) is the application of communication, electronics, machinery, vehicles, control, information, and other technologies to the transportation industry, and the ability to correctly, flexibly, and rapidly understand and propose solutions to maximize the efficiency of transportation and improve traffic conditions [35], [36], [37]. Intelligent transportation can provide better transportation information services for the public, enhance the safety and environmental protection of transport, improve transportation efficiency, and provide more effective means for transportation management [38].

The development of Urban Intelligent Transportation Systems (UITS) went through the ITS 1.0 era in 2005, the ITS 2.0 era in 2015, and the current ITS 3.0 era. It is reconstructs the system service concept and mode with the new technological transformation. Urban traffic managers have started to use big data to drive the innovation and upgrading of traffic management and services. Intelligent travel, intelligent logistics, intelligent assisted driving, etc., have become intelligent transportation's innovation and development direction [39]. Many developed countries, including the United States, Japan, and Singapore, are vigorously developing automatic driving and vehicle networking technology and improving the automation and intelligence of urban traffic management through investment in the construction of intelligent transportation facilities [40].

In the cities of developed countries in Europe and the United States, intelligent transportation is the construction content of general concern. Its core is mainly reflected in the construction of four systems, as follows and shown in Table 1, intelligent perception (such as an intelligent road), intelligent decision-making (such as intelligent logistics, intelligent parking, customized freight), intelligent operation (such as intelligent traffic control) and intelligent services (full-chain personalized services, such as intellectual connection, customized information, and robot pickup) [30]. This also represents the inevitable direction of future urban intelligent transportation development in developing countries.

At present, when the Intelligent Transportation Systems (ITS) in most developing countries cannot fully realize

the intelligent transportation brain and intelligent management system to take over the urban traffic in all directions automatically, traffic managers still need to carry out auxiliary management of urban traffic, and even occupy the main decision-making position [41]. For example, the urban intelligent transportation data mainly relies on the relevant businesses of the transportation bureau and the traffic police bureau, lacks the cross-department, cross-industry data sharing, aggregation of mobile phone signaling data, meteorological data, infrastructure data, railway data, etc., and is difficult to grasp the urban population distribution, land development, urban space, travel rules, etc., which is insufficient to support the traffic management business [42]. Therefore, the gradual transformation of Urban Intelligent Transportation Systems (UITS) in developing countries from semi-automatic to automatic still has a long way to go.

III. INTELLIGENT TRANSPORTATION SYSTEMS IN CHINA

The Urban Intelligent Transportation Systems (UITS) in developing countries and developed countries are quite different and start relatively late [43]. China is a typical representative country of developing countries with a large population, and the management of Urban Intelligent Transportation Systems (UITS) is complex. At present, China has invested more resources in constructing Urban Intelligent Transportation Systems (UITS), and Intelligent Transportation Systems (ITS) have entered a period of rapid development. China's Urban Intelligent Transportation Systems (UITS) is in the stage of system integration, and the whole is in an era of equal emphasis on construction and management [44]. The following describes the Intelligent Transportation Systems of Guangzhou, Shenzhen, and Foshan in China.

A. GUANGZHOU INTELLIGENT TRANSPORTATION SYSTEMS

Guangzhou is the south gate of China. It is an essential central city, international trade center, and comprehensive transportation hub in China, approved by the State Council [45].

At present, facing the needs of urban transport development in the new era, Guangzhou has improved the management system and mechanism of transportation informatization construction and has deeply studied the big data system of urban intelligent transportation. It has built the framework of "one center, three platforms, and four guarantees" intelligent transportation big data system in Guangzhou, continuously promoted the integration and application of new generation information technologies such as big data in the field of urban transportation, and carried out a series of meaningful practices in data perception, processing, application, etc., providing some experience for reference for the application of big data in the transportation industry [46].

According to the policies of the "14th Five Year Plan" for the Reform and Construction of the Guangzhou Digital Government, Guangzhou has built a "one center, three platforms, and four guarantees" [47], [48] urban intelligent

Project	New York	Chicago	San Francisco	Amster- dam	London	Florence
Intelli- gent Parking	✓	✓	✓	✓	✓	\checkmark
Intelli- gent Connec- tion	~	✓	~	~	✓	
Intelli- gent Traffic Infor- mation	~	~	~	~	~	~
Intelli- gent Traffic Light Control System	✓	✓	✓	✓	✓	
Adaptive Commu- nication Connec- tion Vehicle	V		✓			
Share Driver- less Cars	✓		✓	✓	✓	~
Intelli- gent Street Light	✓	✓	✓	✓	✓	~
Ibeacon Location Infor- mation Service		V	~	~	~	~
Custom- ized Freight	~	\checkmark	~	~	\checkmark	~
Robot Pickup	\checkmark	\checkmark	~		\checkmark	
Intelli- gent Logistics	~	~	~	~	~	~

 TABLE 1. Key points of urban intelligent transportation systems construction in developed countries.

transportation big data system, it includes: promoting data perception and sharing, strengthening data transmission efficiency, deepening data mining analysis, and innovating data application services. Among them, "one center" refers to the transportation big data center, and "three platforms" are the intelligent perception platform, integrated business platform, and innovative service platform. Under the intelligent transportation big data system framework, Guangzhou strengthened data fusion analysis and association mining, built cross-dimensional data analysis capabilities, and drove industry governance and service quality improvement with big data innovation. The following introduces several representative system platforms based on the framework structure of the Guangzhou Intelligent Transportation Systems (ITS) [44]. Figure 3 shows the architecture of Guangzhou Intelligent Transportation Systems [48].



FIGURE 3. The architecture of the guangzhou intelligent transportation systems.

1) Guangzhou transportation industry data sharing and analysis service platform. As the core component of the transport big data center, the Guangzhou transport industry data sharing and analysis service platform integrates the data resources of the transport industry in Guangzhou. The data resources cover 17 sectors, more than 800 types of data, and are connected to more than 90 transportation industry information systems. The amount of new data is more than over 250 GB. At the same time, the platform has realized cross-department, cross-industry, cross-field information sharing and access to Guangzhou Public Security Bureau, Metro Group, Meteorological Bureau, Environmental Protection Bureau, and other data resources. It supports passenger flow monitoring and analysis, transportation guarantee, traffic management, expressway and expressway in key areas, road traffic security, and other applications. In addition, the platform has established data standards and specifications, a data resource directory, and data resources of Guangzhou transportation industry core systems such as sharing and basic application services have been sorted and completed by relevant data standards of the Ministry of Transport local data application needs. More than 5000 data standards have further strengthened and standardized the sharing and management of traffic information data and managed the number of transportation industries in the city. According to resources, it provides comprehensive data integration and analysis capabilities for various business systems.

2) Comprehensive monitoring and integrated management platform for traffic operation. The platform adopts the multi-dimensional integrated data fusion management mode to carry out the integrated management of the airport, port, railway, bus, taxi, subway, water bus, passenger, freight transportation, maintenance, driving training, transportation network, station yard, real-time road conditions,



FIGURE 4. Intelligent optimization system for signal timing at road intersections in huangpu district, guangzhou.

crowd passenger flow, and other fields. It realizes the functions of comprehensive information monitoring, multidimensional thematic analysis, early warning, and warning [49], [50], [51]. Through business integration analysis, the platform targeted to carry out decision-making research on multiple business data such as comprehensive transport, public transport, road transport, urban traffic governance, established one-stop management and control of hierarchical, and hierarchical traffic situations to provide support for comprehensive control of traffic situation and scientific command and dispatching.

3) The urban traffic big data system strengthens the threedimensional perception of traffic demand, supply, and the environment by collecting more fine-grained data, especially the quantitative perception of the space-time state of passenger flow and traffic flow. At present, the transportation industry in Guangzhou has formed a multi-source data acquisition mode. Through video, mobile signaling, machine vision, satellite positioning, all-in-one card, sensor, microwave, next-generation Bluetooth and other acquisition channels, it can sense the status information of vehicle flow, passenger flow, logistics, roads, traffic events, and other elements. Among them, the satellite positioning data is 1.25 billion pieces/day, which has fully covered more than 15000 buses, more than 20000 taxis, more than 130000 passenger and freight operating vehicles, and more than 20000 online rental vehicles in Guangzhou, realizing the functions of real-time vehicle positioning, operation safety supervision, travel information services, etc.; There are more than 10000 videos and 20 million business data per day; IC card data is about 9 million pieces/day, Figure 4 shows an intelligent optimization system for signal timing at road intersections in Huangpu District, Guangzhou.

4) Guangzhou Smart China Spring Festival Transport 3.0 system. To relieve the high pressure of passenger flow surge during holidays in China, Guangzhou Transportation Bureau organized the transportation and investment companies affiliated with the public transport group to integrate more than 230 types of data related to transportation, such as public transport, taxi, passenger transport, subway, aviation, railway, video, weather, etc., and upgraded the Smart Spring Festival Transport 3.0 system using big data, artificial intelligence, and other technologies. The system comprehensively and timely grasps the current situation, accurately and efficiently predicts the future trend, quantificationally and directionally dispatches the transport capacity, pre-assesses the impact of measures, and comprehensively evaluates and summarizes, forming a new mode of transportation guarantee for the science and technology spring festival transport.

5) Guangzhou Public Transport Intelligent Management Service Platform. In combination with the establishment of "China's Public Transport City," Guangzhou has promoted the construction of the public transport intelligent application demonstration project and built the Guangzhou public transport intelligent management service platform based on the intelligent public transport management system. The platform carries out macro management from the five sectors of the public transport network, transport capacity, passenger flow, service supervision, and operation monitoring. It carries out comprehensive real-time monitoring of the four major types of characteristic public transport in Guangzhou, including BRT, customized public transport, water bus, and sightseeing loop line, to meet the needs of people for comprehensive query and trend analysis of more than 100 items of information in the public transport industry, with mobile internet The national one-card interconnection and other technologies provide convenient travel services for citizens, integrates face recognition, and intelligent scheduling technology to provide one-click call and timely response services.

6) Guangzhou Urban Comprehensive Traffic Operation Situation Monitoring and Analysis System. Using mobile internet, big data, cloud computing, satellite positioning, and other information technologies based on the data sharing and analysis service platform of the transportation industry, Guangzhou has built a comprehensive urban traffic operation situation monitoring and analysis system that integrates public transport, taxi, passenger transport, road transport, station hub, highway, and other industry supervision services, to grasp the operation of various industries in real-time The rapid development of traffic command and decision-making provides strong technical support.

7) Urban public travel information service platform. To make general travel predictable, Guangzhou has built a comprehensive, multi-mode and all-coverage transportation service system by integrating data from multiple departments, launched a comprehensive transportation information service platform such as "TravelSky," and provided more than 20 one-stop information queries such as travel planning, route navigation, arrival time prediction, waiting for the reminder, and bus arrival reminder, meeting the public's real-time understanding of the location and road conditions on the way Anticipated demands such as travel time, estimated arrival time, etc. In addition, to expand the coverage of information services, the Guangzhou Municipal Bureau of Communications and Transportation has carried out transportation through the mainstream Internet, such as Gaode Software



FIGURE 5. Shenzhen intelligent transportation systems.

Co., Ltd., and Alibaba Group Business cooperation provides more comprehensive and accurate information services for the public.

B. SHENZHEN INTELLIGENT TRANSPORTATION SYSTEMS

Shenzhen is the first special economic zone established by China reform and opening up, the window of China's reform and opening up, and has developed into an influential international city [52], [53]. Shenzhen's Intelligent Transportation Systems is divided into the following three steps, and Figure 5 shows the architecture of Shenzhen Intelligent Transportation Systems.

1) Shenzhen is the first city in China to establish an online transportation system [52]. It has built a public information platform for intelligent transportation, made a breakthrough in transforming the traditional transportation planning process, and formed a quantitative transportation operation forecast and auxiliary decision support system. Shenzhen has determined the three-layer framework of an Intelligent Transportation System (ITS) around information infrastructure, public information platform, and application services. According to the integrated service function of urban traffic planning construction management, 49 user services are provided to management organizations, transport practitioners, product (service) providers, and relevant departments in 9 service areas, including traffic information collection, public information platform, government decision support, traffic information service, public transport management, road operation management, hub logistics information, traffic safety, and vehicle safety, and ITS data management.

2) Taking the opportunity of holding the 26th Universiade, Shenzhen decided to build a "1+3" intelligent system framework for the near future implementation, that is, to build a traffic information sharing platform for the traffic bureau, planning bureau, and the competent departments of the public security traffic police bureau, to build an Intelligent Transportation Systems, traffic planning management, decision support system, and intelligent transportation management system respectively. With its own business, Shenzhen Transport Commission has built a platform (traffic information exchange platform) and five effective systems (traffic integrated monitoring system, traffic integrated control system, traffic operation command system, public travel information service system, and traffic management system). It has accumulated experience in big data collection and integration, application, and new technology storage. The municipal traffic police have basically completed the construction of eight management and control business systems (trunk traffic guidance, parking path guidance, traffic accidents, accident detection and release guidance, illegal management, illegal real-time notification, police service management, and 122 alarm reception and handling), and has initially carried out the collection of basic data, initially can control traffic signals, and has initially realized the basic application of data and multi-channel information release.

3) The development of transportation and management in Shenzhen needs to strengthen the comprehensive connection with the big data platform at the national level. At the same time, the continuous emergence of new forms of transportation, such as online buses, online taxis, and shared bicycles, poses new challenges to the existing traffic management and service system. In 2016, to coordinate the strategic development plan of building a new smart city, Shenzhen successively launched the overall planning of Shenzhen's comprehensive transportation big data platform and the overall improvement project of intelligent transportation control. The former integrates cross-industry and cross-department transportation big-data resources. It promotes the comprehensive and in-depth application of big data in transportation planning, construction, and operation management by following the requirements of "multiple integrations, real-time simulation, and decision support." It also supports Shenzhen's urban transportation planning and construction decisionmaking with big data technology services to increase efficiency and improve comprehensive transportation services and the quality of the refined transportation environment. Build the platform as the decision-making support brain of the city's comprehensive transportation application, and boost the supply-side reform of Shenzhen's transportation infrastructure. The latter aims at active control focus on data fusion and vehicle-road coordination and takes data-driven policing as the direction to transform, expand and integrate the existing system and establish an operation control and policing control system for the whole network, the whole area, and the entire point.

C. FOSHAN INTELLIGENT TRANSPORTATION SYSTEMS

Since implementing the last round of intelligent transportation development planning, Foshan Intelligent Transportation

Systems construction has made specific achievements. In the past few years, Foshan initiated the preparation of the Foshan Intelligent Transport Development Plan (2015-2025) [54], which defined the goal of "full system coordination, high data sharing, and comprehensive service delivery" between the horizontal traffic and traffic police departments, and established the overall framework of "1+2+2", namely, a data exchange and sharing platform, two centers (the municipal public security traffic navigation and control information release center and the municipal comprehensive traffic coordination command center). Two comprehensive applications (comprehensive transportation information service platform and comprehensive transportation decision support system).Foshan intelligent transportation has been constructed in many applications. Figure 5 shows the architecture of Foshan Intelligent Transportation Systems.

1) Regarding industry governance, Foshan's competent department of transportation has successfully launched the construction of intelligent projects related to eight systems, including a taxi service and management information system and a comprehensive management system of vehicles.

2) Regarding traffic system management, it has realized urban traffic management functions such as traffic information collection, monitoring, and information release for the central metropolitan area, trunk roads, and the first ring expressway. It has realized networking and sharing with the city's security video checkpoint system.

3) In the aspect of intelligent public transport, it has built an online intelligent public transport platform across the network, completed the joint network control of the municipal and district governments, TC institutions, and business operations, and realized the automation and intelligence of operation scheduling, cost accounting, industry supervision, decision analysis, and comprehensive services.

4) Regarding travel services, the Municipal Transportation Bureau has successfully completed the construction of relevant projects of "Foshan Transportation" to serve citizens, providing real-time and accurate travel service information for citizens.

5) In terms of transportation decision-making, the construction of the Foshan Transportation Decision Support System was launched in 2012. At present, the basic framework of the system has been built, and the data-sharing access and processing of all departments need to be further deepened.

6) In terms of intelligent transportation data, the "cloud" platform for intelligent transportation virtualization has been built. The data center has gathered 38 application systems, 56 categories and 161 thematic data, forming about 900T of massive traffic data.

D. DEFECTS OF INTELLIGENT TRANSPORTATION SYSTEMS IN THREE CITIES

Intelligent transportation construction has achieved some positive results in China's three ultra-high-density large cities. However, there are still areas for improvement, such as tight constraints on transportation facilities and resources. At present, the proportion of transportation facilities has reached the upper limit of the national standard.

The contradiction between the supply and demand of traffic is increasingly intensifying, and traffic congestion is facing normalization. The development of transportation in the three places has entered a stage of "equal emphasis on construction and management" or even "managementbased." In this case, it is urgent to fully tap the service capacity of existing transportation facilities with the help of new intelligent transportation technology and bring about the improvement of traffic efficiency [55], [56], [57]. In addition, the traffic safety situation is still grim, which is far from advanced international cities. Compared with the death rate per 10000 vehicles, for example, Shenzhen (1.35 in 2015) is currently equivalent to Hong Kong (1.85 in 2014), which is generally equivalent to the level of developed countries in the 1990s. It is the best among the first-tier cities in China, but it is far from the wor- ld-class advanced cities of London (0.45 in 2013), Tokyo (0.38 in 2013), and Berlin (0.28 in 2013) [52]. The new Intelligent Transportation Systems (ITS) will improve traffic safety and the environment and produce substantial social benefits [58]. For example, the new Internet of Vehicles System will reduce the accident rate by 44%, and the New Driverless System will reduce the accidents caused by human factors by 80%. The reliability of traffic information service is insufficient, and the travel experience needs to be improved. Under the new generation of information technology environment, high-precision maps, 5G communications, Internet big data, etc., have laid a strong foundation for improving the travel experience in the whole process, and the means of transportation information services are becoming more and more abundant. However, because the traditional existing information release still relies on the traditional three screens (mobile phone screen, TV screen, and induction screen), there are deficiencies in the diversification, real-time, personalization, and reliability of information services, which need to be improved urgently [59].

More importantly, in the current hierarchical distribution of the three urban intelligent transportation data networks, each organization independently manages and uploads data, which makes it challenging to realize data sharing. Enterprises, government departments, industrial organizations, and relevant scientific research institutions collect static data and build basic databases for business needs [60]. These organizations have limited data collection channels and are limited by their data management capabilities, resulting in imperfect basic database construction and low data collection rate. In contrast, dynamic data is collected by various devices arranged in the Urban Intelligent Transportation Systems(UTIS) [61], [62], [63]. The collected dynamic data directly belongs to the collection equipment management or user organization. In the traditional mode, data belong to various organizations, making verifying and sharing data difficult. Due to the actual demand of the industry for big data sharing, big data com- panies, relevant scientific research

institutions, government statistical departments, and other organizations collect basic data for big data analysis or trading. These comprehensive data management organizations are also organizationally independent, and there are few big data sharing at the same level. Instead, they actively expand the basic data source and collect data from the lower basic data layer [64].

Traffic is chain management, where data is constantly exchanged to improve management [65], [66]. Intelligent traffic is information management based on traffic network infrastructure and traffic flow, which is naturally consistent with the chain mode of blockchain. Introducing blockchain technology into the field of intelligent transportation, carrying out the construction of the underlying platform of transportation blockchain, and breaking through the thinking of internal research in the field of transport are of great significance to mining the potential value of various types of data across areas, industries, and regions [67], [68], [69].

IV. URBAN INTELLIGENT TRANSPORTATION BIG DATA PLATFORM BASED ON BLOCKCHAIN TECHNOLOGY

A. APPLICABILITY AND FEASIBILITY OF BLOCKCHAIN TECHNOLOGY IN URBAN INTELLIGENT TRANSPORTATION SYSTEMS IN DEVELOPING COUNTRIES

Seven significant areas of focus for constructing new infrastructure in developing countries: 5G base stations, ultra-high voltage, intercity high-speed railways, intercity rail transit, new energy vehicle charging piles, big data centers, artificial intelligence, and industrial internet. The essence of new infrastructure is to promote the construction of smart city infrastructure with information digitization infrastructure, improve the importance and security of data, and complement the development of blockchain technology. Artificial intelligence promotes single-point intelligence, industrial internet promotes single-point intelligence integration, and blockchain forms a group contract to establish a trusted mechanism for edge blocks. With edge computing as the core, combined with artificial intelligence and 5G technology [81], the smart road, by deploying smart street lights, establishes a holographic road network system, dynamically tracks and counts block traffic, links traffic trends, and controls intersection information control units in real-time. New infrastructure policies in developing countries provide reliable support for blockchain technology linkage.

Advanced chips at home and abroad continue to be upgraded and supported. High computing power, outdoor process power chips, including industrial grade TX2 and Shengteng 310, support high requirements such as edge artificial intelligence. Disk storage specifications have also made some notable progress in recent years. Industrial-grade storage chips with high capacity and high-speed read and write performance are improving. The needs of various scenarios are gradually sorted out, and business agreements are increasingly improving, coinciding with the digitization of transportation. With the policy support of networking, the

r feasibility of blockchain landing in the transportation industry is growing. Under the pressure load of the new cloud center, the development characteristics of blockchain technology with edge computing as the core meet the requirements of the era of intelligent transportation.

Relying on the decentralized, tamper-proof, and traceable characteristics of blockchain technology, taking block data as the core, removing the centralized data management of various organizations, changing the data collection, data processing and analysis, data storage mode, methods, and fully realizing the platform-based big data sharing, decentralized and distributed computing of the multi-source system of urban intelligent transportation. Using blockchain technology in big data can prevent data from being arbitrarily added, modified, and deleted, greatly enhancing the elasticity and storage capacity of big data. While enhancing security, various technologies in big data can be interacted with without barriers, forming a higher level of interaction and generating new data support.

Applying blockchain technology to intelligent transportation big data platforms can solve various problems of existing big data centers. Firstly, blockchain technology's distributed and decentralized nature distributes hundreds of servers and massive storage across the nodes of each participant, which can significantly reduce the system's deployment, operation, and maintenance costs. Secondly, blockchain technology adopts the concept of peer-to-peer nodes. In addition to a unified basic service node, each participant node also focuses on its business. On the one hand, it has dramatically reduced the difficulty of technological research and development of the unified transportation big data platform. On the other hand, it has improved the processing capacity of dedicated data. Finally, the introduction of smart contracts allows the system to complete the introduction of new business models and data models only through software deployment in the form of smart contracts, without the need to reconstruct the storage structure of the back-end database and the data structure of the access interface, significantly improving the flexibility and scalability of the system.

B. BIG DATA PLATFORM ARCHITECTURE OF URBAN INTELLIGENT TRANSPORTATION BASED ON BLOCKCHAIN TECHNOLOGY

The urban intelligent transportation data block in the data layer includes static and dynamic data blocks. Both types of data have timestamp characteristics [70]. The static data is relatively poor in liquidity, and the dynamic data is more timevarying. Therefore, these two data types are constructed separately in the blockchain data layer to form an interconnected blockchain network [71]. Moreover, in some cases, urban transportation big data is a combination of dynamic data and static data. Therefore, a consensus algorithm combining dynamic and static traffic data should be established at this layer in the future. Accordingly, the dynamic and static data consensus algorithms are established in the consensus layers, respectively. At the incentive level, the resource provision mechanism and resource adjustment mechanism are used to encourage the data sharing of all nodes and realize the network's decentralization, as shown in Table 2.

The basic architecture of the big data platform of urban intelligent transportation based on blockchain technology reconstructs the basic data layer of the traditional urban intelligent transportation data network architecture, excluding the basic data source layer. The relationship between urban intelligent transportation data blocks and data sources, and adds regulatory access servers for each blockchain node to ensure the security of information. It can be seen that the urban intelligent transportation big data platform of blockchain technology takes block data as the core, removes the centralized data management of various organizations, and fully realizes the platform-based big data sharing. Especially through the role of the consensus layer, incentive layer, and contract layer, data can be verified, trusted, interactive, and easy to analyze; Through the network layer and data layer, the data can be easily traced, stored, standardized, unified and standardized, and the data of the urban intelligent transportation data network can be more complete and authentic. The data platform can be more optimized, intelligent, and global.

C. ADVANTAGES OF BLOCKCHAIN TECHNOLOGY IN URBAN INTELLIGENT TRANSPORTATION BIG DATA PLATFORM

Under the traditional intelligent transportation data network architecture, take urban road traffic dynamic information as an example: road congestion information is collected by the data source terminal equipment of the transportation department or uploaded by the user of the navigation software; Relevant government units upload the control information; Road environmental information is mainly collected by monitoring equipment of meteorological department; Vehicle distribution information is provided or collected by navigation platform, operation company, and traffic monitoring equipment [72], [73], [74]. The data sources of this information are different. After each provider of the basic data management layer processes the collected data, the comprehensive data management layer processes the data again. In the whole process, the data is collected in a decentralized way and then processed by the organizations of the basic data management layer, and then collected and processed by the comprehensive data management layer for the second time. Due to the independence of the organizations of the basic data management layer and the comprehensive data management layer, data cannot be shared at the same level but can be transferred between different levels according to the business and functions of each organization.

Under the big data platform architecture of urban intelligent transportation based on blockchain technology, all data sources are directly connected to the network through blockchain nodes, and data processing and analysis are completed by the network itself rather than by various
 TABLE 2. The architecture of big data platform for urban intelligent transportation based on blockchain technology.

Before Optimization	Blockchain Architecture	After Optimization	
Programmable Currency		Industry Users	
Programmable Finance	Application Laver	Government Users	
Programmable Social Industry		Resident users	
Script Code		Script Code	
Algorithm Mechanism	Contract Layer	Algorithm Mechanism	
Smart Contract		Smart Contract	
Issuance Mechanism	Incentive Laver	Resource Provision Mecha- nism	
Allocation Mechanism	incentive Layer	Resource Regulation Mech- anism	
PoW		Dynamic Data Consensus Algorithm	
	Consensus Layer	Static Data Consensus Algorithm	
POS DPOS		Dynamic And Static Combi- nation Algorithm	
P2P Network		P2P Network	
Communication Mechanism	Network Layer	Communication Mechanism	
Verification Mechanism		UIT Data Validation Mecha- nism	
Data Block		Urban Intelligent Transportation system Data Block	
Chain Structure		Chain Structure	
Time Stamp		Trusted Timestamp	
Hash Function	Data Layer	Cryptographic algorithm (Hash algorithm)	
Merkle Tree		Merkle Tree	
Asymmetric Encryption		Asymmetric Encryption	

organizations so that data is "shared" from beginning to end. Moreover, the distributed computing and decentralization of blockchain ensure the "levelization" of all data and avoid the loss, distortion, and verification difficulties of inter-level data transmission.

In comparison, the urban intelligent transportation big data platform architecture of blockchain technology has fundamentally changed the traditional intelligent transportation data network architecture, completely changed the data collection, data processing, analysis, data storage mode, and methods, and truly realized big data sharing, decentralization, and distributed computing. It can be predicted that the big data platform of urban intelligent transportation based on blockchain technology can bring advantages such as significantly reducing system costs, significantly increasing system data volume and encouraging data sharing, removing the limitation of administrative management level, improving computing efficiency and related resource utilization, expanding application fields and service objects, improving data professional analysis ability, and improving system data reliability.

D. KEY TECHNOLOGIES OF BLOCKCHAIN IN BIG DATA PLATFORM FOR URBAN INTELLIGENT TRANSPORTATION

The main problems to be solved and faced by the big data platform of urban intelligent transportation based on blockchain technology are the unification of different data sources, the unified supervision and operation of data, and the technical compatibility with other advanced technologies. To solve these problems, the key technologies of blockchain technology on the big data platform of urban intelligent transportation are discussed from the perspective of technology.

1) Unified technology of each node. The unification of data from different sources includes the unification of data format, data acquisition frequency, data storage structure, data update, modification, etc. The unification of data from other data sources is realized between blocks and data sources, data layer, and network layer [75]. The data sources of urban intelligent transportation data blocks are different. To achieve the unification of data, first of all, through the supervision and access server, the network access technical certification and technical supervision of the data source are carried out to ensure that the data source can technically meet the basic standards of network access, and ensure that the data source can upload data without fault, redundancy, and error during the operation process [76], [77], [78]. After that, in-depth analysis is needed to analyze the relationship between data provided by different data sources, including source relationship, logical relationship, value relationship, relevance relationship, accuracy comparison, etc. Based on this, the urban intelligent transportation data block model with unified data is determined, and the data block is constructed. Finally, the blockchain's data layer and network layer technology are used to realize the unified storage and transmission of data based on the urban intelligent transportation data block.

2) Supervision and operation realization of multiple technology sharing. The big data platform of urban intelligent transportation based on blockchain technology differs from the traditional big data platform of urban intelligent transportation. Different organizations supervise the data of the conventional urban intelligent transportation big data platform and then share it on the platform, which makes it challenging to achieve unified supervision and operation of data.

The decentralization of the big data platform of urban intelligent transportation based on blockchain technology is the "ownership" of data, which enables the unified intelligent management of data after access to the network to achieve suitable data supervision and automatic operation. Specifically, the unified supervision and operation of data is realized at the consensus, incentive, contract, and application layers of the urban intelligent transportation big data platform of blockchain technology and is the result of the sharing of multiple technologies. In terms of data supervision and operation, the consensus layer, the incentive layer, the contract layer, and the application layer have respectively constructed the technical processing mechanism, the resource allocation mechanism, the management rule mechanism, and the application mechanism of data supervision and operation. All data blocks are automatically processed, transmitted, and provided services according to these mechanisms.

3) Blockchain technology is compatible with the Internet of Vehicles and vehicle-road collaboration technology. The Internet of Vehicles is a dynamic mobile communication system that realizes the network communication between vehicles and the public through the interaction between vehicles & vehicles, vehicles & roads, vehicles & people, and vehicles & sensor equipment. In essence, the Internet of Vehicles can also be seen as a networked realization of multi-source data around vehicles, roads, and people [79], [80]. The key issues of the Internet of Vehicles are what data to collect, what network communication technology to use, and what data applications to provide.

Blockchain technology can be directly used for the data network communication of the Internet of Vehicles as the technical basis of the Internet of Vehicles data network. It can also be used as the external network link of the data network communication of the Internet of Vehicles when meeting the technical compatibility requirements to assist or supplement the various functions of the Internet of Vehicles [81], [82], [83], [84]. In addition to realizing the intelligent perception of people, vehicles, and roads, vehicle-road coordination is more important to learn system coordination, optimizing system resource use, and achieving traffic supervision and operation objectives such as improving road traffic safety and easing traffic congestion. In addition to the basic data network functions, blockchain technology's consensus layer and contract layer can be used as the basis to ensure system collaboration. In contrast, the incentive layer can achieve the goal of optimizing the use of system resources.

The sharing of multiple technologies in blockchain technology can realize automatic supervision and operation. Therefore, applying blockchain technology to vehicle-road collaboration can achieve various objectives of the vehicleroad association. The blockchain nodes of the big data platform of urban intelligent transportation based on blockchain technology can theoretically connect different data sources and have technical compatibility [85], [86], [87]. As long as the data source meets each node's unified technical standards and requirements, the data source can be connected to the network by generating a new blockchain node. Therefore, the big data platform of urban intelligent transportation based on blockchain technology is open.

4) Trusted timestamp technology [88]. A trusted timestamp is an electronic certificate issued by the Joint Trust Timestamp Service Center that can prove that a data message (electronic file) exists, is complete, verifiable, and has a legal effect at a certain point in time. Trusted timestamps are mainly used to prevent tampering and subsequent repudiation of electronic files and to determine the exact time when electronic files are generated. According to the International Electronic Timestamp Standard Specification "RFC 3161", the essence of a trusted timestamp service is to bind the hash value of a user's electronic data to an authoritative time source, and on this basis, generate an unforgeable timestamp file through a joint trusted timestamp service center digital signature.

5) Ledger Database [89]. In actual blockchain applications, many scenarios do not have a strong demand for decentralization, but they only want their tamper-proof, traceable, transparent, and auditable features. Based on this, Alibaba Cloud has designed LedgerDB, a centralized ledger database that cannot be tampered with. Ledger DB is more lightweight than the heavy licensing chain and has a high throughput (up to 80 times that of Hyperledger Fabric). It also supports strong auditing, which can simultaneously prevent DB users and service providers from committing crimes to meet the demand for ledger expansion, Ledger DB can delete historical data but still ensure the verifiability of the data. Moreover, it can "modify historical data" (which seems to violate the essence of blockchain but is indeed the requirement of real scenarios). Some already linked data has problems (such as non-compliance), and it needs to be hidden later.

V. CONSTRUCTION CONCEPT OF NEW GENERATION INTELLIGENT TRANSPORTATION SYSTEMS

The new generation of Intelligent Transportation Systems (ITS) is a new type of intelligent transportation ecosystem that is guided by the balance of urban traffic supply and demand, based on the in-depth perception and research, and analysis of urban traffic travel characteristics, realizes the cross-department collaborative decision-making of urban traffic management through advanced technologies such as blockchain and artificial intelligence, and can dynamically optimize the matching of urban traffic supply and traffic demand space-time resources through self-learning and optimization. As a new generation of Intelligent Transportation Systems (ITS), the construction of the traffic brain serves the needs of fully automatic and accurate traffic management is applied to solve the persistent problems of urban traffic, which helps greatly improve the efficiency of traffic operation, and injects new vitality and intelligence into the development of urban traffic industry [90].

Traditional Intelligent Transportation Systems emphasize hardware construction in the development process. At the same time, the system control strategy needs a lot of human resources invested in the implementation process, which can not realize the auxiliary implementation of automation. The lag effect is evident in the long-term urban traffic management process [91]. In the face of the unprecedented travel pressure of the current city and the new opportunities brought by the rise of the new generation of information technology such as blockchain, big data, cloud computing for the development of intelligent transportation, and construction of the new generation of the intelligent transportation system has become an inevitable choice for the development of the times and the construction of smart new forms of urban transportation. It also frees urban planners and managers from heavy business processing and creates immediate control benefits.

The development of blockchain technology is driving ITS towards a new mode of transportation operation, management, and service. In the future, traffic management will be reshaped into AI global decision-making, automatic adjustment of traffic systems based on travelers' needs and preferences, and highly developed mobile internet and intelligence. It can be predicted that the intelligent traffic management system will gradually rely on blockchain technology, big data perception, and artificial intelligence decision-making to provide customized and highly intelligent services for traffic managers and travelers, as follows and shown in Figure 7.

A. KEY FEATURES OF THE NEW GENERATION INTELLIGENT TRANSPORTATION

Intelligent transportation will be based on big data, blockchain, artificial intelligence, and the Internet of Things to build an urban traffic management ecosystem with mobile connectivity, collaborative control & intelligent services and comprehensively improve the traditional Intelligent Transportation Systems [92], [93], [94]. In the future, cities will present the key features of mobile awareness of global information, smart governance of traffic decision-making, and accurate customization of travel services, including training application blockchain AI is the driving force for entering the era of intelligent transportation and industry 4.0 [95].

1) As a complex and multi-dimensional physical network, urban transportation in the future should have the ability to global information perception and gather massive, heterogeneous, and complex big data such as automatic data collection by front-end equipment, manual data collection, administrative data collection, shared data collection, and data provided by the public voluntarily. Utilize advanced block, mobile internet, Internet of Things, and other information technologies to sense the real-time status information of road, vehicle, traveler, facility, and other elements in the massive multiple data, realize cross-level, cross-domain, cross-system, cross-business information exchange, and break the current situation of information management and split operation, realize the interactivity of intelligent transportation information exchange and the universality of information application, for the sake of accuracy Customized traffic management and travel services provide a data awareness platform.

2) In the future, smart transportation will deeply integrate blockchain, artificial intelligence, deep learning, and other technologies with traffic management, with strong cognitive and learning abilities, and be able to continuously selflearning and self-evolution in the process of business processing to adapt to the changing and complex traffic development situation. It has changed the previous empirical decisionmaking mode of solving traffic problems and formed a highly automated command & decision-making system that integrates perception and prediction, decision-making and management, and evaluation services. It can complete the refined perception of the traffic situation and laws, intelligent decision-making of traffic control schemes, accurate analysis



FIGURE 6. Foshan intelligent transportation systems.



FIGURE 7. Top-level architecture of intelligent transportation brain in yun city based on blockchain.

and optimization of traffic services, and liberate the limited police resources investment based on the global shared big data. It will enable the intelligent mode to cover the whole business field of traffic control.

3) Using the intelligent information interconnection and sharing mechanism, intelligent transportation can trace the travelers' personal preferences in the future and provide more personalized directional service information for travelers in combination with real-time perception and prediction of the traffic operation status of the road network. At the same time, we will improve the ability of comprehensive travel information service to cover the whole chain, mode & region, achieve real-time, diversified & accurate information queries, and release and intelligent feedback throughout the process. Make full use of the holographic information perception tentacles of intelligent transportation, integrate travel guidance and traffic police control business, and promote the optimization and improvement of urban, comprehensive decision-making, facility construction, demand regulation, and other aspects based on individual travel behavior.

B. BRAIN STRUCTURE OF FUTURE URBAN INTELLIGENT TRANSPORTATION

With the enrichment and improvement of the vast and complex urban intelligent transportation management system, building the urban intelligent transportation brain will be the trend of intelligent transportation development in a city, especially in a developing country.

Under the background of the rapid development of a new generation of information and intelligence technologies such as big data, blockchain, and artificial intelligence, the urban intelligent transportation brain is to comprehensively acquire, deeply analyze, comprehensively study and judge, intelligently generate countermeasures, precise decision-making, and system application of urban and urban transportationrelated information through comprehensive intelligence such as human-like brain's perception, cognition, coordination, learning, control, decision-making, feedback, and innovation Cyclic optimization is the core hub of Urban Intelligent Transportation Systems(UITS) better to realize the governance and service of urban traffic, solve the problems of urban traffic and provide systematic and comprehensive services. Therefore, its construction needs to follow the top-level scientific, establish hierarchical construction objectives, and achieve various functional requirements with the goal as the guide. Based on the latest technology's application status and development trend, the pilot overall planning of the future traffic management function is carried out.

The urban intelligent transportation brain will form a top-level structure of "one cloud and four centers," integrating the transportation cloud, perception center, analysis center, decision-making center, and evaluation center. Based on blockchain technology, the Brain Traffic Cloud is responsible for interconnecting and sharing all reliable and safe blockchain distribution data provided by the existing business systems of traffic police and other parts and managing and dispatching all business centers. The perception, analysis, decision-making, and evaluation centers are the basic engines for realizing various functions of the intelligent transportation brain. They respectively undertake the important functions of global perception, real-time analysis, intelligent decisionmaking, and scientific evaluation.

The perception center relies on blockchain, big data technology and other holographic perception of urban traffic conditions to realize multi-level, all-time and even nodelevel comprehensive perception of traffic facilities, traffic demand and traffic operation situation; Through blockchain technology, artificial intelligence and machine learning, the analysis center supports the analysis of abnormal condition such as traffic congestion mechanism analysis and judgment, traffic incident research and review, and early warning, and real-time deduces and predicts the space-time passenger flow characteristics and traffic operation status; The decision center builds a closed-loop control mechanism for the coordinated operation of "perception - analysis - decision evaluation" of traffic big data, and provides the formulation, distribution and resource allocation of comprehensive traffic management solutions, as well as visual scheme and wholeprocess tracking and supervision; The evaluation center uses the traffic simulation technology and mathematical model to track the characteristics and evolution trend of traffic operation under the multi-class status quo and optimized traffic control measures, carefully evaluate the implementation plan and identify the existing problems, and support the travel planning and guidance services based on real-time traffic conditions and behavior analysis, intelligent parking guidance services, and the sharing and real-time release of various types of traffic information.

In the future, the urban intelligent traffic brain will realize brain-like reasoning and decision-making to assist urban management, traffic control [96], and public travel services based on blockchain, big data, artificial intelligence, and other new generation information technologies, cultivate the self-learning, self-evolution ability of the traffic brain and continuously improve its intelligence so that it can realize closed-loop intelligent management of autonomous perception, intelligent analysis, intelligent decision-making [97], and scientific evaluation in some application businesses in the long-term stage, It provides comprehensive and intelligent services for the future development of the city, traffic business management and public travel.

VI. FUTURE PROSPECTS

This survey paper has covered blockchain technology and Urban Intelligent Transportation Systems (UITS) integration architecture model in developing countries. Intelligent Transportation Systems (ITS) are the key to improving the efficiency, service quality, safety level, environmental protection, and energy conservation of the transportation system. Relying on the new generation of information technology, such as blockchain and big data, Intelligent Transportation Systems in developing countries will achieve rapid development, and the level of intelligence will be significantly improved. The future development of intelligent transportation will focus on building a sharing platform for urban transportation systems [98], creating an advanced and practical urban "intelligent transportation brain," making a world-leading Urban Intelligent Transportation Systems (UITS), achieving high-level vehicle and road coordination, improving the intelligent level of passenger and freight transportation services [99], achieving intelligent, comprehensive transportation, solving traffic congestion with the help of a high degree of intelligence, improving the safety level, and achieving green traffic dominance.

Prospects of future Urban Intelligent Transportation Systems (UITS) is an urban transportation strategy and countermeasure system composed of urban land use form adjustment, urban transportation structure adjustment, urban road network structure optimization, and urban road system right-of-way adjustment. It should serve urban development planning and comprehensive transportation development planning. To do a good job in the construction of the Urban Intelligent Transportation Systems (UITS), we should attach great importance to the top-level design of the Intelligent Transportation Systems (ITS), adhere to the principle of traffic infrastructure and traffic engineering facilities construction first, not only pay attention to the hardware construction of the Intelligent Transportation Systems (ITS) but also pay attention to the software development and function improvement of the Intelligent Transportation Systems (ITS).

The construction of Urban Intelligent Transportation Systems (UITS) in the future should focus on realizing functions and be problem-oriented [100]. It should be both progressive and practical. The first key to developing an Intelligent Transportation System is to achieve the expected function and actual application effect rather than the system construction itself. The structure of Intelligent Transportation Systems should be fully demonstrated by experts, interviewed by citizens, and analyzed by construction effect. Clear system objectives, clear acceptance criteria, sufficient expert argumentation, and a perfect post-evaluation system are the basic requirements to ensure the healthy development of ITS. We should speed up the introduction of relevant standards and specifications, form a perfect system, guide the construction of an Intelligent Transportation System (ITS), and avoid decision-making mistakes and investment waste.

Although the Urban Intelligent Transportation Systems (UITS) integrated with blockchain technology has a good development prospect, blockchain technology still has some potential security risks. For example: 1) 51% of attacks [101]. When the malicious attack node of the entire network holds 51% of the total computing power, it can attack the whole network and maliciously alter or delete the general ledger. 2) Anonymity issues [102]. Although the cryptography principles used in the current blockchain seem impregnable, if the key is acquired or stolen, no third party can help retrieve it. And with the emergence of quantum computing, the encryption key may be quickly cracked, thus destroying the foundation of the blockchain. 3) Illegal attack [103]. In the blockchain mesh structure, malicious nodes will intercept new blocks' work and transaction processing and even

disconnect the entry registration channel with the network. Malicious nodes may also block the new blocks created by other nodes so that the main chain can only link the blocks created by itself. Therefore, there is still a long way to go to build Urban Intelligent Transportation Systems (UITS) in developing countries in the future.

VII. CONCLUSION

The development of blockchain, big data, artificial intelligence technology, and other technologies has led to the innovation of urban government management, traffic police traffic management, and resident travel service, which can improve more refined and three-dimensional urban services. Developing countries still have a long way to go in constructing Urban Intelligent Transportation Systems (UITS) for Various factors. The paper describes in detail the current situation of the construction of Intelligent Transportation Systems (ITS) in three cities in China, including the research system architecture, functions, functions, advantages, and disadvantages. It proposes a structural model for building a big data platform for Urban Intelligent Transportation Systems (UITS) based on blockchain technology. This paper also constructs a toplevel design structure model of an Urban Intelligent Transportation System (UITS) based on blockchain traffic cloud, which can also be used for reference by developing countries. A detailed discussion concerning future and open research avenues is also performed, which could help pave the way for researchers and urban transport system designers to explore the key challenging areas in integrating blockchain technology and Urban Intelligent Transportation Systems (UITS), especially in developing countries.

REFERENCES

- B. Mikavica and A. Kostić-Ljubisavljević, "Blockchain-based solutions for security, privacy, and trust management in vehicular networks: A survey," J. Supercomput., vol. 77, pp. 1–56, Feb. 2021.
- [2] S. Nakamoto, "Bitcoin: A peer-to-peer electronic cash system," White Paper, Nov. 2008. [Online]. Available: https://bitcoin.org/bitcoin.pdf
- [3] N. Khoshavi, G. Tristani, and A. Sargolzaei, "Blockchain applications to improve operation and security of transportation systems: A survey," *Electronics*, vol. 10, no. 5, p. 629, Mar. 2021.
- [4] M. Swan, "The chapter of preface," in *Blockchain: Blueprin for a New Economy*. Sebastopol, CA, USA: O'Reilly Media, 2015, p. 9.
- [5] B. Pillai, K. Biswas, Z. Hou, and V. Muthukkumarasamy, "The burn-toclaim cross-blockchain asset transfer protocol," in *Proc. 25th Int. Conf. Eng. Complex Comput. Syst. (ICECCS)*, Oct. 2020, pp. 119–124.
- [6] K. J. Button, P. Nijkamp, and P. Rietveld, *Transport—Maker and Breaker of Cities*, C. Clark, Ed. Northampton, MA, USA: Urban Transport, Edward Elgar Publishers, 2003, pp. 503–516.
- [7] A. Alvarez-Risco, S. Del-Aguila-Arcentales, and M. A. Rosen, *Sustainable Transportation in Cities*. Springer, 2020, pp. 149–165, doi: 10.1007/978-3-030-45533-0_12.
- [8] H. H. Jeong, Y. C. Shen, J. P. Jeong, and T. T. Oh, "A comprehensive survey on vehicular networking for safe and efficient driving in smart transportation: A focus on systems, protocols, and applications," *Veh. Commun.*, vol. 31, Oct. 2021, Art. no. 100349, doi: 10.1016/j.vehcom.2021.100349.
- [9] L. Zhu, F. R. Yu, Y. Wang, B. Ning, and T. Tang, "Big data analytics in intelligent transportation systems: A survey," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 1, pp. 383–398, Jan. 2019.
- [10] Z. Zhou, M. Wang, J. Huang, S. Lin, and Z. Lv, "Blockchain in big data security for intelligent transportation with 6G," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 7, pp. 9736–9746, Jul. 2022, doi: 10.1109/TITS.2021.3107011.

- [11] A. Lamssaggad, N. Benamar, A. S. Hafid, and M. Msahli, "A survey on the current security landscape of intelligent transportation systems," *IEEE Access*, vol. 9, pp. 9180–9208, 2021.
- [12] L. Qi, "Research on intelligent transportation systems technologies and applications," in *Proc. Workshop Power Electron. Intell. Transp. Syst.*, 2008, pp. 529–531.
- [13] S.-H. An, B.-H. Lee, and D.-R. Shin, "A survey of intelligent transportation systems," in *Proc. IEEE 3rd Int. Conf. Comput. Intell., Commun. Syst. Netw.*, Jul. 2011, pp. 332–337.
- [14] N.-E.-E. Faouzi, H. Leung, and A. Kurian, "Data fusion in intelligent transportation systems: Progress and challenges—A survey," *Inf. Fusion*, vol. 12, no. 1, pp. 4–10, Jan. 2011.
- [15] J. Zhang, F.-Y. Wang, K. Wang, W.-H. Lin, X. Xu, and C. Chen, "Datadriven intelligent transportation systems: A survey," *IEEE Trans. Intell. Transp. Syst.*, vol. 12, no. 4, pp. 1624–1639, Dec. 2011.
- [16] G. Dimitrakopoulos and P. Demestichas, "Intelligent transportation systems," *IEEE Veh. Technol. Mag.*, vol. 5, no. 1, pp. 77–84, Mar. 2010.
- [17] K. Zheng, Q. Zheng, P. Chatzimisios, W. Xiang, and Y. Zhou, "Heterogeneous vehicular networking: A survey on architecture, challenges, and solutions," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 4, pp. 2377–2396, 4th Quart., 2015.
- [18] T. A. Butt, R. Iqbal, K. Salah, M. Aloqaily, and Y. Jararweh, "Privacy management in social Internet of Vehicles: Review, challenges and blockchain based solutions," *IEEE Access*, vol. 7, pp. 79694–79713, 2019.
- [19] V. Astarita, V. P. Giofrè, G. Mirabelli, and V. Solina, "A review of blockchain-based systems in transportation," *Information*, vol. 11, no. 1, p. 21, Dec. 2019.
- [20] M. T. Çaldağ and E. Gökalp, "Exploring critical success factors for blockchain-based intelligent transportation systems," *Emerg. Sci. J.*, vol. 4, pp. 27–44, Oct. 2020.
- [21] A. Beikverdi and J. Song, "Trend of centralization in bitcoin's distributed network," in *Proc. ACIS Int. Conf. Softw. Eng. Artif. Intell. Netw. Parallel/Distrib. Comput.* Piscataway, NJ, USA: IEEE Compfer Society, 2015, pp. 1–6.
- [22] A. Sarker, S. Byun, W. Fan, and S.-Y. Chang, "Blockchain-based root of trust management in security credential management system for vehicular communications," in *Proc. 36th Annu. ACM Symp. Appl. Comput.*, Mar. 2021, pp. 223–231.
- [23] D. Moussaoui, B. Kadri, M. Feham, and B. A. Bensaber, "A distributed blockchain based PKI (BCPKI) architecture to enhance privacy in VANET," in *Proc. 2nd Int. Workshop Hum.-Centric Smart Environ. Health Well-Being (IHSH)*, Feb. 2021, pp. 75–79.
- [24] M. S. Ali, M. Vecchio, M. Pincheira, K. Dolui, F. Antonelli, and M. H. Rehmani, "Applications of blockchains in the Internet of Things: A comprehensive survey," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 2, pp. 1676–1717, 2nd Quart., 2019.
- [25] H. Wang, Z. Zheng, S. Xie, H. N. Dai, and X. Chen, "Blockchain challenges and opportunities: A survey," *Int. J. Web Grid Services*, vol. 14, no. 4, p. 352, 2018, doi: 10.1504/IJWGS.2018.10016848.
- [26] W. Wang, D. T. Hoang, P. Hu, Z. Xiong, D. Niyato, P. Wang, Y. Wen, and D. I. Kim, "A survey on consensus mechanisms and mining strategy management in blockchain networks," *IEEE Access*, vol. 7, pp. 22328–22370, 2019, doi: 10.1109/ACCESS.2019.2896108.
- [27] T. Neudecker and H. Hartenstein, "Network layer aspects of permissionless blockchains," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 1, pp. 838–857, 1st Quart., 2018.
- [28] M. Andoni, V. Robu, D. Flynn, S. Abram, D. Geach, D. Jenkins, P. McCallum, and A. Peacock, "Blockchain technology in the energy sector: A systematic review of challenges and opportunities," *Renew. Sustain Energy Rev.*, vol. 100, pp. 143–174, Feb. 2019.
- [29] M. Pournader, Y. Shi, S. Seuring, and S. C. L. Koh, "Blockchain applications in supply chains, transport and logistics: A systematic review of the literature," *Int. J. Prod. Res.*, vol. 58, no. 7, pp. 2063–2081, Apr. 2020.
- [30] X. Zhang, Y. Shao, and C. Sun, "The overall concept of smart transportation for future cities," *J. Urban Transp.*, vol. 10, no. 5, pp. 1–7, Sep. 2018.
- [31] R. Soltani, M. Zaman, R. Joshi, and S. Sampalli, "Distributed ledger technologies and their applications: A review," *Appl. Sci.*, vol. 12, no. 5, pp. 1–20, 2022, doi: 10.3390/app12157898.
- [32] M. A. Khan and K. Salah, "IoT security: Review, blockchain solutions, and open challenges," *Future Gener. Comput. Syst.*, vol. 82, pp. 395–411, May 2018.

- [33] X. Li, P. Jiang, T. Chen, X. Luo, and Q. Wen, "A survey on the security of blockchain systems," *Future Gener. Comput. Syst.*, vol. 107, pp. 841–853, Jun. 2020, doi: 10.1016/j.future.2017.08.020.
- [34] S. Ghaemi, S. Rouhani, R. Belchior, R. S. Cruz, H. Khazaei, and P. Musilek, "A pub-sub architecture to promote blockchain interoperability," 2021, arXiv:2101.12331.
- [35] Y. Yuan and F.-Y. Wang, "Towards blockchain-based intelligent transportation systems," in *Proc. IEEE 19th Int. Conf. Intell. Transp. Syst.* (*ITSC*), Nov. 2016, pp. 2663–2668.
- [36] Y. Li, K. Ouyang, N. Li, R. Rahmani, H. Yang, and Y. Pei, "A blockchainassisted intelligent transportation system promoting data services with privacy protection," *Sensors*, vol. 20, no. 9, p. 2483, Apr. 2020.
- [37] S. Sharma and B. Kaushik, "A survey on Internet of Vehicles: Applications, security issues & solutions," Veh. Commun., vol. 20, Dec. 2019, Art. no. 100182.
- [38] Y. Dai, D. Xu, S. Maharjan, G. Qiao, and Y. Zhang, "Artificial intelligence empowered edge computing and caching for Internet of Vehicles," *IEEE Wireless Commun.*, vol. 26, no. 3, pp. 12–18, Jun. 2019.
- [39] L. Mendiboure, M. A. Chalouf, and F. Krief, "Survey on blockchainbased applications in Internet of Vehicles," *Comput. Electr. Eng.*, vol. 84, Jun. 2020, Art. no. 106646.
- [40] A. S. Kulathunge and H. R. O. E. Dayarathna, "Communication framework for vehicular ad-hoc networks using blockchain: Case study of metro Manila electric shuttle automation project," in *Proc. Int. Res. Conf. Smart Comput. Syst. Eng. (SCSE)*, Mar. 2019, pp. 85–90.
- [41] H. Liao, Y. Mu, Z. Zhou, M. Sun, Z. Wang, and C. Pan, "Blockchain and learning-based secure and intelligent task offloading for vehicular fog computing," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 7, pp. 4051–4063, Jul. 2021.
- [42] G. T. S. Ho, Y. P. Tsang, C. H. Wu, W. H. Wong, and K. L. Choy, "A computer vision-based roadside occupation surveillance system for intelligent transport in smart cities," *Sensors*, vol. 19, no. 8, p. 1796, Apr. 2019.
- [43] F. Golbabaei, T. Yigitcanlar, and J. Bunker, "The role of shared autonomous vehicle systems in delivering smart urban mobility: A systematic review of the literature," *Int. J. Sustain. Transp.*, vol. 15, pp. 1–18, Jul. 2020, doi: 10.1080/15568318.2020.1798571.
- [44] Z. wu, X. Wu, and L. wang, "Prospects for the development trend of smart transportation under the background of powerful transportation," *Transp. Res.*, vol. 5, no. 4, pp. 26–36, 2019.
- [45] L. J. Chen, "Preliminary analysis on the Guangzhou City image in the bargeboards of Pearl River and Guangzhou City scenery map in Qing dynasty," J. Voprosy IstorII, vol. 3, no. 12, pp. 310–317, 2022.
- [46] G. Ke, "Research on the integration of comprehensive transportation hub and urban transportation system—A case of Guangzhou north railway station," *City Building*, vol. 16, no. 313, pp. 28–29, 2019.
- [47] S. Guo, "Multiple projects have achieved national innovation! Guangzhou will create a '2.0 version of urban intelligent transportation systems," Yangcheng Evening News All Media, Aug. 25, 2022. [Online]. Available: https://www.163.com/dy/article/HFKNU FNV0550AXYG.html
- [48] Z. Zhang, Q. Huang, and C. Feng, "Research and practice on Guangzhou urban intelligent transportation big data system," *J. Big Data*, vol. 4, no. 5, Jul. 2019, pp. 113–120. [Online]. Available: https://www.sohu.com/a/342191621_389742
- [49] S. K. Singh, J. H. Park, P. K. Sharma, and Y. Pan, "BIIoVT: Blockchainbased secure storage architecture for intelligent internet of vehicular things," *IEEE Consum. Electron. Mag.*, vol. 11, no. 6, pp. 75–82, Nov. 2022, doi: 10.1109/MCE.2021.3089992.
- [50] Q. Wang, L. Zhou, Z. Tang, and G. Wang, "A consortium blockchainbased model for data sharing in Internet of Vehicles," in *Proc. Int. Conf. Smart City Informatization.* Singapore: Springer, 2019, pp. 253–267.
- [51] A. Patel, N. Shah, T. Limbasiya, and D. Das, "VehicleChain: Blockchainbased vehicular data transmission scheme for smart city," in *Proc. IEEE Int. Conf. Syst., Man Cybern. (SMC)*, Oct. 2019, pp. 661–667.
- [52] J. H. Song, X. Wang, Z. Jin, and X. L. Tang, "Shenzhen smart transportation innovation and practice in the background of urban traffic quality improvement," *J. Transp.*, vol. 6, no. 32, Jul. 2019, pp. 148–153.
- [53] G. Qin and L. Liang, "Overview of quality transportation and its practice in Shenzhen: A case study of street quality improvement of Futian CBD, Shenzhen," J. Urban Transp., vol. 2, no. 17, pp. 54–61, Feb. 2019.
- [54] H. W. Qing, Y. Yongjie, Z. Chaohua, and W. W. Fan, "Study on the development direction of intelligent transportation systems of Foshan City under big data background," in *Proc. 15th China Intell. Transp. Annu. Conf.*, Nov. 2020, pp. 1–8, doi: 10.26914/c.cnkihy.2020.028425.

- [55] S. Islam, S. Badsha, S. Sengupta, H. La, I. Khalil, and M. Atiquzzaman, "Blockchain-enabled intelligent vehicular edge computing," *IEEE Netw.*, vol. 35, no. 3, pp. 125–131, May 2021.
- [56] Y. Zhang, J. Misic, and Z. Zheng, "Guest editorial introduction to the special section on blockchain for vehicles and intelligent communications," *IEEE Trans. Veh. Technol.*, vol. 70, no. 5, pp. 3998–4000, May 2021.
- [57] A. Vangala, B. Bera, S. Saha, A. K. Das, N. Kumar, and Y. Park, "Blockchain-enabled certificate-based authentication for vehicle accident detection and notification in intelligent transportation systems," *IEEE Sensors J.*, vol. 21, no. 14, pp. 15824–15838, Jul. 2021.
- [58] H. Lu, "Progress of intelligent transportation systems key technologies," J. Sci. Technol. Guide, vol. 6, no. 37, pp. 27–35, Mar. 2019.
- [59] M. B. Mollah, J. Zhao, D. Niyato, Y. L. Guan, and L. H. Koh, "Blockchain for the Internet of Vehicles towards intelligent transportation systems: A survey," *IEEE Internet Things J.*, vol. 8, no. 6, pp. 4157–4185, Mar. 2021.
- [60] Y. Li, K. Ouyang, N. Li, R. Rahmani, H. Yang, and Y. Pei, "A blockchainassisted intelligent transportation system promoting data services with privacy protection," *Sensors*, vol. 20, no. 9, p. 2483, Apr. 2020.
- [61] D. Wang and X. Zhang, "Secure data sharing and customized services for intelligent transportation based on a consortium blockchain," *IEEE Access*, vol. 8, pp. 56045–56059, 2020.
- [62] A. M. Krishna and A. K. Tyagi, "Intrusion detection in intelligent transportation system and its applications using blockchain technology," in *Proc. Int. Conf. Emerg. Trends Inf. Technol. Eng. (ic-ETITE)*, Feb. 2020, pp. 1–8.
- [63] T. Zhou, J. Shen, Y. Ren, and S. Ji, "Threshold key management scheme for blockchain-based intelligent transportation systems," *Secur. Commun. Netw.*, vol. 2021, pp. 1–8, Sep. 2021.
- [64] N. Deepa, Q.-V. Pham, D. C. Nguyen, S. Bhattacharya, B. Prabadevi, T. R. Gadekallu, P. K. R. Maddikunta, F. Fang, and P. N. Pathirana, "A survey on blockchain for big data: Approaches, opportunities, and future directions," *Future Gener. Comput. Syst.*, vol. 131, pp. 209–226, Jun. 2022.
- [65] Z. Li, Z. Xiao, Q. Xu, E. Sotthiwat, R. S. M. Goh, and X. Liang, "Blockchain and IoT data analytics for fine-grained transportation insurance," in *Proc. IEEE 24th Int. Conf. Parallel Distrib. Syst. (ICPADS)*, Dec. 2018, pp. 1022–1027.
- [66] L. Benarous, B. Kadri, and A. Bouridane, "Blockchain-based privacyaware pseudonym management framework for vehicular networks," *Arabian J. Sci. Eng.*, vol. 45, no. 8, pp. 1–17, 2020.
- [67] L.-A. Hîrtan, C. Dobre, and H. González-Vélez, "Blockchain-based reputation for intelligent transportation systems," *Sensors*, vol. 20, no. 3, p. 791, 2020.
- [68] R. Chaudhary, A. Jindal, G. S. Aujla, S. Aggarwal, N. Kumar, and K.-K.-R. Choo, "BEST: Blockchain-based secure energy trading in SDNenabled intelligent transportation system," *Comput. Secur.*, vol. 85, pp. 288–299, Aug. 2019.
- [69] R. Kumar, P. Kumar, R. Tripathi, G. P. Gupta, N. Kumar, and M. M. Hassan, "A privacy-preserving-based secure framework using blockchain-enabled deep-learning in cooperative intelligent transport system," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 9, pp. 16492–16503, Sep. 2022, doi: 10.1109/TITS.2021.3098636.
- [70] X. Wang, C. Xu, Z. Zhou, S. Yang, and L. Sun, "A survey of blockchainbased cybersecurity for vehicular networks," in *Proc. Int. Wireless Commun. Mobile Comput. (IWCMC)*, Jun. 2020, pp. 740–745.
- [71] H. Ye and S. Park, "Reliable vehicle data storage using blockchain and IPFS," *Electronics*, vol. 10, no. 10, p. 1130, May 2021.
- [72] M. Kamal, G. Srivastava, and M. Tariq, "Blockchain-based lightweight and secured V2V communication in the Internet of Vehicles," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 7, pp. 3997–4004, Jul. 2021.
- [73] Q. Kong, L. Su, and M. Ma, "Achieving privacy-preserving and verifiable data sharing in vehicular fog with blockchain," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 8, pp. 4889–4898, Aug. 2021.
- [74] K. M. Vidya, R. Koduri, S. Nandyala, and M. Manalikandy, "Secure vehicular communication using blockchain technology," SAE Tech. Paper 2020-01-0722, 2020.
- [75] B. Li, R. Liang, D. Zhu, W. Chen, and Q. Lin, "Blockchain-based trust management model for location privacy preserving in VANET," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 6, pp. 3765–3775, Jun. 2021.
- [76] T. Salman, M. Zolanvari, A. Erbad, R. Jain, and M. Samaka, "Security services using blockchains: A state of the art survey," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 1 pp. 858–880, 1st Quart., 2019, doi: 10.1109/COMST.2018.2863956.

- [77] G. Wang, "SoK: Exploring blockchains interoperability," IACR Cryptol. ePrint Arch., vol. 2021, p. 537, Dec. 2021.
- [78] B. Pillai, K. Biswas, Z. Hóu, and V. Muthukkumarasamy, "Burn-to-claim: An asset transfer protocol for blockchain interoperability," *Comput. Netw.*, vol. 200, Dec. 2021, Art. no. 108495.
- [79] Z. Zheng, J. Pan, and L. Cai, "Lightweight blockchain consensus protocols for vehicular social networks," *IEEE Trans. Veh. Technol.*, vol. 69, no. 6, pp. 5736–5748, Jun. 2020.
- [80] R. Gupta, A. Kumari, and S. Tanwar, "A taxonomy of blockchain envisioned edge-as-a-connected autonomous vehicles," *Trans. Emerg. Telecommun. Technol.*, vol. 32, no. 6, p. e4009, Jun. 2021.
- [81] S. Wang, S. Sun, X. Wang, Z. Ning, and J. J. P. C. Rodrigues, "Secure crowdsensing in 5G Internet of Vehicles: When deep reinforcement learning meets blockchain," *IEEE Consum. Electron. Mag.*, vol. 10, no. 5, pp. 72–81, Sep. 2021.
- [82] A. Bonadio, F. Chiti, R. Fantacci, and V. Vespri, "An integrated framework for blockchain inspired fog communications and computing in Internet of Vehicles," *J. Ambient Intell. Humanized Comput.*, vol. 11, no. 2, pp. 755–762, Feb. 2020.
- [83] P. Bagga, A. K. Sutrala, A. K. Das, and P. Vijayakumar, "Blockchain-based batch authentication protocol for Internet of Vehicles," *J. Syst. Archit.*, vol. 113, Feb. 2021, Art. no. 101877.
- [84] U. Javaid and B. Sikdar, "A secure and scalable framework for blockchain based edge computation offloading in social Internet of Vehicles," *IEEE Trans. Veh. Technol.*, vol. 70, no. 5, pp. 4022–4036, May 2021.
- [85] S. Distefano, A. D. Giacomo, and M. Mazzara, "Trustworthiness for transportation ecosystems: The blockchain vehicle information system," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 4, pp. 2013–2022, Apr. 2021.
- [86] K. Abbas, L. A. Tawalbeh, A. Rafiq, A. Muthanna, I. A. Elgendy, and A. A. A. El-Latif, "Convergence of blockchain and IoT for secure transportation systems in smart cities," *Secur. Commun. Netw.*, vol. 2021, pp. 1–13, Apr. 2021.
- [87] J. Zhao, C. He, C. Peng, and X. Zhang, "Blockchain for effective renewable energy management in the intelligent transportation system," *J. Interconnection Netw.*, vol. 22, Mar. 2022, Art. no. 2141009.
- [88] X. Yang, S. Wang, F. Li, Y. Zhang, W. Yan, F. Gai, B. Yu, L. Feng, Q. Gao, and Y. Li, "Ubiquitous verification in centralized ledger database," in *Proc. IEEE 38th Int. Conf. Data Eng. (ICDE)*, May 2022, pp. 1821–1880, doi: 10.1109/ICDE53745.2022.00181.
- [89] X. Yang, Y. Zhang, S. Wang, B. Yu, F. Li, and Y. Li, "LedgerDB: A centralized ledger database for universal audit and verification," *Proc. VLDB Endowment*, vol. 13, no. 12, pp. 3138–3151, 2020. [Online]. Available: https://www.jianshu.com/p/1c8de5e63c10doi: 10.14778/3415478.3415540
- [90] S. Bao, Y. Cao, A. Lei, P. Asuquo, H. Cruickshank, Z. Sun, and M. Huth, "Pseudonym management through blockchain: Cost-efficient privacy preservation on intelligent transportation systems," *IEEE Access*, vol. 7, pp. 80390–80403, 2019.
- [91] D. Zhao, N. Thakur, and J. Chen, "Optimal design of energy storage system to buffer charging infrastructure in smart cities," *J. Manage. Eng.*, vol. 36, no. 2, Mar. 2020, Art. no. 04019048.
- [92] H.-N. Dai, Z. Zheng, and Y. Zhang, "Blockchain for Internet of Things: A survey," *IEEE Internet Things J.*, vol. 6, no. 5, pp. 8076–8094, Oct. 2019.
- [93] T. Reimers, F. Leber, and U. Lechner, "Integration of blockchain and Internet of Things in a car supply chain," in *Proc. IEEE Int. Conf. Decentralized Appl. Infrastruct. (DAPPCON)*, Apr. 2019, pp. 146–151.
- [94] S. Megha, H. Salem, E. Ayan, and M. Mazzara, "A survey of blockchain solutions for autonomous vehicles ecosystems," *J. Phys., Conf. Ser.*, vol. 1694, no. 1, Dec. 2020, Art. no. 012024.
- [95] H. Chauhan, D. Kumar, D. Gupta, S. Gupta, and V. Verma, "Blockchain and IoT based vehicle tracking system for industry 4.0 applications," *IOP Conf. Ser., Mater. Sci. Eng.*, vol. 1022, no. 1, Jan. 2021, Art. no. 012051.
- [96] Q. Wang, T. Ji, Y. Guo, L. Yu, X. Chen, and P. Li, "TrafficChain: A blockchain-based secure and privacy-preserving traffic map," *IEEE Access*, vol. 8, pp. 60598–60612, 2020.

- [97] Y. Ren, F. Zhu, J. Wang, P. K. Sharma, and U. Ghosh, "Novel vote scheme for decision-making feedback based on blockchain in Internet of Vehicles," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 2, pp. 1639–1648, Feb. 2022.
- [98] L. F. Luque-Vega, M. A. Carlos-Mancilla, V. G. Payán-Quiñónez, and E. Lopez-Neri, "Smart cities oriented project planning and evaluation methodology driven by citizen perception—IoT smart mobility case," *Sustainability*, vol. 12, no. 17, p. 7088, Aug. 2020.
- [99] C. Oham, R. A. Michelin, R. Jurdak, S. S. Kanhere, and S. Jha, "B-FERL: Blockchain based framework for securing smart vehicles," *Inf. Process. Manage.*, vol. 58, no. 1, Jan. 2021, Art. no. 102426.
- [100] X. Lin, J. Wu, S. Mumtaz, S. Garg, J. Li, and M. Guizani, "Blockchainbased on-demand computing resource trading in IoV-assisted smart city," *IEEE Trans. Emerg. Topics Comput.*, vol. 9, no. 3, pp. 1373–1385, Feb. 2020. [Online]. Available: https://Ieeex-plore.ieee.org/abstract/ document/8985250?casa_token=MycqpviqK0UAAAAA:tzXslCnroqKCG C61pcUWcK5WxBCQpWvS5lkrYBqy7kbieyoz3QefO7nS7nl-JmO_ ZO20c64YYXA
- [101] R. Jabbar, E. Dhib, A. B. Said, M. Krichen, N. Fetais, E. Zaidan, and A. K. Barkaoui, "Blockchain technology for intelligent transportation systems: A systematic literature review," *IEEE Access*, vol. 10, pp. 2169–3536, 2022.
- [102] M. N. M. Bhutta, A. A. Khwaja, A.-N. H. F. Ahmad, M. K. Khan, M. A. Hanif, H. Song, M. Alshamari, and Y. Cao, "A survey on blockchain technology: Evolution, architecture and security," *IEEE Access*, vol. 9, pp. 2169–3536, 2021.
- [103] S. Sayeed and H. Marco-Gisbert, "Proof of adjourn (PoAj): A novel approach to mitigate blockchain attacks," *Appl. Sci.*, vol. 10, no. 18, p. 6607, Sep. 2020.



XINGCHENG GUO (Member, IEEE) received the master's degree from Sun Yat-sen University, Guangzhou, China. He is currently an Engineer with the China's Ministry of Industry and Information Technology and a Lecturer with the Guangzhou City University of Technology. He is selected as one of the first batches of the top 100 E-Business teachers in Guangdong Province, China. He is an active researcher with many papers at national and international. He has

contributed to several national and provincial research projects, including those funded by the Guangdong Provincial University Committee and the Guangdong Higher Education Association. His research interests include smart cities, intelligent transportation, blockchain, the development of developing countries, E-business, and internet security.



XIANGLONG GUO is currently the Director of the Industrial Development Strategy Research Institute, Nanchang, China. He is an economist certified by the Chinese Government. He has more than 30 years of research experience in business operations and engineering technology. He is a consultant to the local government and many enterprises. His research interests include the development of developing countries, intelligent transportation, blockchain, and internet security.

•••